Author's response to reviews

Title: Identification of ambiguities in the 1994 chronic fatigue syndrome research case definition and recommendations for resolution

Authors:

Bart Stouten (bartstouten@wanadoo.nl)

Version: 3 Date: 12 April 2005

Author's response to reviews:

Dear Editors,

I herewith submit the final version of my manuscript. Below is a point-to-point description of the changes made:

- removed the article type from the title page
- provided a structured abstract (Background, Methods, Results, Conclusions)
- removed the word 'Sir' from the manuscript text
- structured and rearranged the manuscript sections
- incorporated the appendix in the main text
- adjusted the numbering of the references (the order of appearance changed now that the appendix is incorporated in the main text)
- corrected "which happens when $\bar{Y}$ is less than $S_{m-1}$" to "which happens when $\bar{Y}$ is less than $NS_{m-1}$"
- changed "for proofreading the manuscript" in the acknowledgement to "for proofreading the original manuscript"

Yours faithfully,
Bart Stouten